Author's response to reviews

Title: Household Exposure to Violence and Human Rights Violations in Western Bangladesh (I): Prevalence, Risk Factors and Consequences

Authors:

Shr-Jie Wang (sjw@rct.dk)
Jens Modvig (jmo@rct.dk)
Edith Montgomery (em@rct.dk)

Version: 6 Date: 2 November 2009

Author's response to reviews:

We have now revised the manuscript according to the suggestions of the editors for both major revisions and minor revisions. We confirm that all authors have read and approved the final version of the manuscript. Overall title for figure 3 is provided and the legends of figures 3a, 3b and 3c are included in the figures. We had used sentence case in the title and formatted the abbreviations as a paragraph. All the tables fit into A4 pages the table title is only one sentence long. Additional information was treated as a legend and it appears below the table. We have also removed the visible vertical lines from all the tables. We have corrected the typing errors.